

GLOBAL MASTER SECURITIES LENDING AGREEMENT

CONTENTS

CLAU	P	AGE
1.	APPLICABILITY	3
2.	INTERPRETATION	3
3.	LOANS OF SECURITIES	8
4.	DELIVERY	8
5.	COLLATERAL	9
6.	DISTRIBUTIONS AND CORPORATE ACTIONS	13
7.	RATES APPLICABLE TO LOANED SECURITIES AND CASH COLLATERA	AL14
8.	DELIVERY OF EQUIVALENT SECURITIES	15
9.	FAILURE TO DELIVER	16
10.	EVENTS OF DEFAULT	17
11.	CONSEQUENCES OF AN EVENT OF DEFAULT	18
12.	TAXES	21
13.	LENDER'S WARRANTIES	23
14.	BORROWER'S WARRANTIES	23
15.	INTEREST ON OUTSTANDING PAYMENTS	23
16.	TERMINATION OF THIS AGREEMENT	24
17.	SINGLE AGREEMENT	24
18.	SEVERANCE	24
19.	SPECIFIC PERFORMANCE	24
20.	NOTICES	24
21.	ASSIGNMENT	25
22.	NON-WAIVER	25
23.	GOVERNING LAW AND JURISDICTION	25
24.	TIME	25
25.	RECORDING	25
26.	WAIVER OF IMMUNITY	25
27.	MISCELLANEOUS	26
CCHEI	OUT E	20

AGR	EEMEN	NT DATED AS OF20						
BETV	VEEN:							
	the la	ai Junan Securities (Hong Kong) Limited (Part was of Hong Kong OF 27/F, Low Block, Grand M ral, Hong Kong acting through one or more Designa	illennium Plaza	, 181 (
			(Party	\boldsymbol{B})	holder	of		
		number	, who	se prir	ncipal plac	e of		
	reside	ence is						
	Party	A is licensed and registered with:-						
	ca fi ac	ecurities and Futures Commission as a Licensed Control of the regulated activities of (1) dealing an annual accommodation to facilitate acquisitions or dvising on securities; and (3) dealing in futures control of the activities as an issuer of products listed	in securities (in holdings of sec entracts (only fo	cludin urities r hedg	g provision by Client ing purpo	on of (); (2) () se in		
	(b) T	The Stock Exchange of Hong Kong Limited as an Ex	change Particip	ant.				
1.	APPL	LICABILITY						
1.1	transa finand with a Secur	time to time the Parties acting through one or more actions in which one party (<i>Lender</i>) will transfer to cial instruments (<i>Securities</i>) against the transfer of Ca simultaneous agreement by Borrower to transfer to rities on a fixed date or on demand against the transfer to such Collateral.	the other (<i>Borr</i> Collateral (as def Lender Securitie	ower) ined in es equi	securities paragraph valent to s	and h 2) such		
1.2	by the the So In the Agree	such transaction shall be referred to in this Agreeme terms of this Agreement, including the supplement chedule and any Addenda or Annexes attached hereto e event of any inconsistency between the provisions ement, the provisions of such Addendum or Annexes agree.	tal terms and cor o, unless otherwi of an Addendu	ndition ise agre m or A	s containe eed in writ nnex and	ed in ting. this		
1.3	Eithe: Nomi	r Party may perform its obligations under this Agrinee.	reement either d	lirectly	or throug	gh a		
2.	INTE	Interpretation						
2.1	In thi	In this Agreement:						
	Act o	f Insolvency means in relation to either Party:						
	(a)	its making a general assignment for the benefit of arrangement, or composition with creditors; or	of, or entering in	nto a r	eorganisat	tion,		

its stating in writing that it is unable to pay its debts as they become due; or

(b)

- (c) its seeking, consenting to or acquiescing in the appointment of any trustee, administrator, receiver or liquidator or analogous officer of it or any material part of its property; or
- (d) the presentation or filing of a petition in respect of it (other than by the other Party to this Agreement in respect of any obligation under this Agreement) in any court or before any agency alleging or for the bankruptcy, winding-up or insolvency of such Party (or any analogous proceeding) or seeking any reorganisation, arrangement, composition, re-adjustment, administration, liquidation, dissolution or similar relief under any present or future statute, law or regulation, such petition not having been stayed or dismissed within 30 days of its filing (except in the case of a petition for winding-up or any analogous proceeding in respect of which no such 30 day period shall apply); or
- (e) the appointment of a receiver, administrator, liquidator or trustee or analogous officer of such Party over all or any material part of such Party's property; or
- (f) the convening of any meeting of its creditors for the purpose of considering a voluntary arrangement as referred to in Section 3 of the Insolvency Act 1986 (or any analogous proceeding);

Agency Annex means the Annex to this Agreement published by the International Securities Lending Association and providing for Lender to act as agent for a third party in respect of one or more Loans;

Alternative Collateral means Collateral having a Market Value equal to the Collateral delivered pursuant to paragraph 5 and provided by way of substitution in accordance with the provisions of paragraph 5.3;

Applicable Law means the laws, rules and regulations (including double taxation conventions) of any relevant jurisdiction, including published practice of any government or other taxing authority in connection with such laws, rules and regulations;

Automatic Early Termination has the meaning given in paragraph 10.1(d);

Base Currency means the currency indicated in paragraph 2 of the Schedule;

Business Day means:

- (a) in relation to Delivery in respect of any Loan, a day other than a Saturday or a Sunday on which banks and securities markets are open for business generally in the place(s) where the relevant Securities, Equivalent Securities, Collateral or Equivalent Collateral are to be delivered;
- (b) in relation to any payments under this Agreement, a day other than a Saturday or a Sunday on which banks are open for business generally in the principal financial centre of the country of which the currency in which the payment is denominated is the official currency and, if different, in the place where any account designated by the Parties for the making or receipt of the payment is situated (or, in the case of a payment in euro, a day on which TARGET operates);
- (c) in relation to a notice or other communication served under this Agreement, any day other than a Saturday or a Sunday on which banks are open for business generally in the place designated for delivery in accordance with paragraph 3 of the Schedule; and

in any other case, a day other than a Saturday or a Sunday on which banks are open for business generally in each place stated in paragraph 6 of the Schedule;

Buy-In means any arrangement under which, in the event of a seller or transferor failing to deliver securities to the buyer or transferee, the buyer or transferee of such securities is entitled under the terms of such arrangement to buy or otherwise acquire securities equivalent to such securities and to recover the cost of so doing from the seller or transferor;

Cash Collateral means Collateral taking the form of a transfer of currency;

Close of Business means the time at which the relevant banks, securities settlement systems or depositaries close in the business centre in which payment is to be made or Securities or Collateral is to be delivered;

Collateral means such securities or financial instruments or transfers of currency as are referred to in the table set out under paragraph 1 of the Schedule as being acceptable or any combination thereof as agreed between the Parties in relation to any particular Loan and which are delivered by Borrower to Lender in accordance with this Agreement and shall include Alternative Collateral;

Defaulting Party has the meaning given in paragraph 10;

Delivery in relation to any Securities or Collateral or Equivalent Securities or Equivalent Collateral comprising Securities means:

- (a) in the case of Securities held by a Nominee or within a clearing or settlement system, the crediting of such Securities to an account of the Borrower or Lender, as the case may be, or as it shall direct, or,
- (b) in the case of Securities otherwise held, the delivery to Borrower or Lender, as the case may be, or as the transferee shall direct of the relevant instruments of transfer, or
- (c) by such other means as may be agreed,

and *deliver* shall be construed accordingly;

Designated Office means the branch or office of a Party which is specified as such in paragraph 6 of the Schedule or such other branch or office as may be agreed to in writing by the Parties;

Equivalent or equivalent to in relation to any Loaned Securities or Collateral (whether Cash Collateral or Non-Cash Collateral) provided under this Agreement means Securities or other property, of an identical type, nominal value, description and amount to particular Loaned Securities or Collateral (as the case may be) so provided. If and to the extent that such Loaned Securities or Collateral (as the case may be) consists of Securities that are partly paid or have been converted, subdivided, consolidated, made the subject of a takeover, rights of pre-emption, rights to receive securities or a certificate which may at a future date be exchanged for Securities, the expression shall include such Securities or other assets to which Lender or Borrower (as the case may be) is entitled following the occurrence of the relevant event, and, if appropriate, the giving of the relevant notice in accordance with paragraph 6.7 and provided that Lender or Borrower (as the case may be) has paid to the other Party all and any sums due in respect thereof. In the event that such Loaned Securities or Collateral (as the case may be) have been redeemed, are partly paid, are the subject of a capitalisation issue or are subject to an event similar to any of the foregoing events described in this paragraph, the expression shall have the following meanings:

- (a) in the case of redemption, a sum of money equivalent to the proceeds of the redemption;
- (b) in the case of a call on partly-paid Securities, Securities equivalent to the relevant Loaned Securities or Collateral, as the case may be, provided that Lender shall have paid Borrower, in respect of Loaned Securities, and Borrower shall have paid to Lender, in respect of Collateral, an amount of money equal to the sum due in respect of the call;
- (c) in the case of a capitalisation issue, Securities equivalent to the relevant Loaned Securities or Collateral, as the case may be, together with the securities allotted by way of bonus thereon;
- (d) in the case of any event similar to any of the foregoing events described in this paragraph, Securities equivalent to the Loaned Securities or the relevant Collateral, as the case may be, together with or replaced by a sum of money or Securities or other property equivalent to that received in respect of such Loaned Securities or Collateral, as the case may be, resulting from such event;

Income means any interest, dividends or other distributions of any kind whatsoever with respect to any Securities or Collateral;

Income Record Date, with respect to any Securities or Collateral, means the date by reference to which holders of such Securities or Collateral are identified as being entitled to payment of Income:

Letter of Credit means an irrevocable, non-negotiable letter of credit in a form, and from a bank, acceptable to Lender;

Loaned Securities means Securities which are the subject of an outstanding Loan;

Margin has the meaning specified in paragraph 1 of the Schedule with reference to the table set out therein:

Market Value means:

- (a) in relation to the valuation of Securities, Equivalent Securities, Collateral or Equivalent Collateral (other than Cash Collateral or a Letter of Credit):
 - (i) such price as is equal to the market quotation for the mid price of such Securities, Equivalent Securities, Collateral and/or Equivalent Collateral as derived from a reputable pricing information service reasonably chosen in good faith by Lender; or
 - (ii) if unavailable the market value thereof as derived from the mid price or rate bid by a reputable dealer for the relevant instrument reasonably chosen in good faith by Lender,

in each case at Close of Business on the previous Business Day, or as specified in the Schedule, unless agreed otherwise or, at the option of either Party where in its reasonable opinion there has been an exceptional movement in the price of the asset in question since such time, the latest available price, plus (in each case):

(iii) the aggregate amount of Income which has accrued but not yet been paid in respect of the Securities, Equivalent Securities, Collateral or Equivalent Collateral concerned to the extent not included in such price,

provided that the price of Securities, Equivalent Securities, Collateral or Equivalent Collateral that are suspended or that cannot legally be transferred or that are transferred or required to be transferred to a government, trustee or third party (whether by reason of nationalisation, expropriation or otherwise) shall for all purposes be a commercially reasonable price agreed between the Parties, or absent agreement, be a price provided by a third party dealer agreed between the Parties, or if the Parties do not agree a third party dealer then a price based on quotations provided by the Reference Dealers. If more than three quotations are provided, the Market Value will be the arithmetic mean of the prices, without regard to the quotations having the highest and lowest prices. If three quotations are provided, the Market Value will be the quotation remaining after disregarding the highest and lowest quotations. For this purpose, if more than one quotation has the same highest or lowest price, then one of such quotations shall be disregarded. If fewer than three quotations are provided, the Market Value of the relevant Securities, Equivalent Securities, Collateral or Equivalent Collateral shall be determined by the Party making the determination of Market Value acting reasonably;

- (b) in relation to a Letter of Credit the face or stated amount of such Letter of Credit; and
- (c) in relation to Cash Collateral the amount of the currency concerned;

Nominee means a nominee or agent appointed by either Party to accept delivery of, hold or deliver Securities, Equivalent Securities, Collateral and/or Equivalent Collateral or to receive or make payments on its behalf;

Non-Cash Collateral means Collateral other than Cash Collateral;

Non-Defaulting Party has the meaning given in paragraph 10;

Notification Time means the time specified in paragraph 1.5 of the Schedule;

Parties means Lender and Borrower and **Party** shall be construed accordingly;

Posted Collateral has the meaning given in paragraph 5.4;

Reference Dealers means, in relation to any Securities, Equivalent Securities, Collateral or Equivalent Collateral, four leading dealers in the relevant securities selected by the Party making the determination of Market Value in good faith;

Required Collateral Value has the meaning given in paragraph 5.4;

Sales Tax means value added tax and any other Tax of a similar nature (including, without limitation, any sales tax of any relevant jurisdiction);

Settlement Date means the date upon which Securities are due to be transferred to Borrower in accordance with this Agreement;

Stamp Tax means any stamp, transfer, registration, documentation or similar Tax; and

Tax means any present or future tax, levy, impost, duty, charge, assessment or fee of any nature (including interest, penalties and additions thereto) imposed by any government or

other taxing authority in respect of any transaction effected pursuant to or contemplated by, or any payment under or in respect of, this Agreement.

2.2 **Headings**

All headings appear for convenience only and shall not affect the interpretation of this Agreement.

2.3 Market terminology

Notwithstanding the use of expressions such as "borrow", "lend", "Collateral", "Margin" etc. which are used to reflect terminology used in the market for transactions of the kind provided for in this Agreement, title to Securities "borrowed" or "lent" and "Collateral" provided in accordance with this Agreement shall pass from one Party to another as provided for in this Agreement, the Party obtaining such title being obliged to deliver Equivalent Securities or Equivalent Collateral as the case may be.

2.4 Currency conversions

Subject to paragraph 11, for the purposes of determining any prices, sums or values (including Market Value and Required Collateral Value) prices, sums or values stated in currencies other than the Base Currency shall be converted into the Base Currency at the latest available spot rate of exchange quoted by a bank selected by Lender (or if an Event of Default has occurred in relation to Lender, by Borrower) in the London inter-bank market for the purchase of the Base Currency with the currency concerned on the day on which the calculation is to be made or, if that day is not a Business Day, the spot rate of exchange quoted at Close of Business on the immediately preceding Business Day on which such a quotation was available.

2.5 The Parties confirm that introduction of and/or substitution (in place of an existing currency) of a new currency as the lawful currency of a country shall not have the effect of altering, or discharging, or excusing performance under, any term of the Agreement or any Loan thereunder, nor give a Party the right unilaterally to alter or terminate the Agreement or any Loan thereunder. Securities will for the purposes of this Agreement be regarded as equivalent to other securities notwithstanding that as a result of such introduction and/or substitution those securities have been redenominated into the new currency or the nominal value of the securities has changed in connection with such redenomination.

2.6 Modifications etc. to legislation

Any reference in this Agreement to an act, regulation or other legislation shall include a reference to any statutory modification or re-enactment thereof for the time being in force.

3. LOANS OF SECURITIES

Lender will lend Securities to Borrower, and Borrower will borrow Securities from Lender in accordance with the terms and conditions of this Agreement. The terms of each Loan shall be agreed prior to the commencement of the relevant Loan either orally or in writing (including any agreed form of electronic communication) and confirmed in such form and on such basis as shall be agreed between the Parties. Unless otherwise agreed, any confirmation produced by a Party shall not supersede or prevail over the prior oral, written or electronic communication (as the case may be).

4. **DELIVERY**

4.1 Delivery of Securities on commencement of Loan

Lender shall procure the Delivery of Securities to Borrower or deliver such Securities in accordance with this Agreement and the terms of the relevant Loan.

4.2 Requirements to effect Delivery

The Parties shall execute and deliver all necessary documents and give all necessary instructions to procure that all right, title and interest in:

- (a) any Securities borrowed pursuant to paragraph 3;
- (b) any Equivalent Securities delivered pursuant to paragraph 8;
- (c) any Collateral delivered pursuant to paragraph 5;
- (d) any Equivalent Collateral delivered pursuant to paragraphs 5 or 8;

shall pass from one Party to the other subject to the terms and conditions set out in this Agreement, on delivery of the same in accordance with this Agreement with full title guarantee, free from all liens, charges and encumbrances. In the case of Securities, Collateral, Equivalent Securities or Equivalent Collateral title to which is registered in a computer-based system which provides for the recording and transfer of title to the same by way of book entries, delivery and transfer of title shall take place in accordance with the rules and procedures of such system as in force from time to time. The Party acquiring such right, title and interest shall have no obligation to return or deliver any of the assets so acquired but, in so far as any Securities are borrowed by or any Collateral is delivered to such Party, such Party shall be obliged, subject to the terms of this Agreement, to deliver Equivalent Securities or Equivalent Collateral as appropriate.

4.3 Deliveries to be simultaneous unless otherwise agreed

Where under the terms of this Agreement a Party is not obliged to make a Delivery unless simultaneously a Delivery is made to it, subject to and without prejudice to its rights under paragraph 8.6, such Party may from time to time in accordance with market practice and in recognition of the practical difficulties in arranging simultaneous delivery of Securities, Collateral and cash transfers, waive its right under this Agreement in respect of simultaneous delivery and/or payment provided that no such waiver (whether by course of conduct or otherwise) in respect of one transaction shall bind it in respect of any other transaction.

4.4 **Deliveries of Income**

In respect of Income being paid in relation to any Loaned Securities or Collateral, Borrower (in the case of Income being paid in respect of Loaned Securities) and Lender (in the case of Income being paid in respect of Collateral) shall provide to the other Party, as the case may be, any endorsements or assignments as shall be customary and appropriate to effect, in accordance with paragraph 6, the payment or delivery of money or property in respect of such Income to Lender, irrespective of whether Borrower received such endorsements or assignments in respect of any Loaned Securities, or to Borrower, irrespective of whether Lender received such endorsements or assignments in respect of any Collateral.

5. COLLATERAL

5.1 Delivery of Collateral on commencement of Loan

Subject to the other provisions of this paragraph 5, Borrower undertakes to deliver to or deposit with Lender (or in accordance with Lender's instructions) Collateral simultaneously with

Delivery of the Securities to which the Loan relates and in any event no later than Close of Business on the Settlement Date.

5.2 Deliveries through securities settlement systems generating automatic payments

Unless otherwise agreed between the Parties, where any Securities, Equivalent Securities, Collateral or Equivalent Collateral (in the form of securities) are transferred through a book entry transfer or settlement system which automatically generates a payment or delivery, or obligation to pay or deliver, against the transfer of such securities, then:

- (a) such automatically generated payment, delivery or obligation shall be treated as a payment or delivery by the transferee to the transferor, and except to the extent that it is applied to discharge an obligation of the transferee to effect payment or delivery, such payment or delivery, or obligation to pay or deliver, shall be deemed to be a transfer of Collateral or delivery of Equivalent Collateral, as the case may be, made by the transferee until such time as the Collateral or Equivalent Collateral is substituted with other Collateral or Equivalent Collateral if an obligation to deliver other Collateral or deliver Equivalent Collateral existed immediately prior to the transfer of Securities, Equivalent Securities, Collateral or Equivalent Collateral; and
- (b) the Party receiving such substituted Collateral or Equivalent Collateral, or if no obligation to deliver other Collateral or redeliver Equivalent Collateral existed immediately prior to the transfer of Securities, Equivalent Securities, Collateral or Equivalent Collateral, the Party receiving the deemed transfer of Collateral or Delivery of Equivalent Collateral, as the case may be, shall cause to be made to the other Party for value the same day either, where such transfer is a payment, an irrevocable payment in the amount of such transfer or, where such transfer is a Delivery, an irrevocable Delivery of securities (or other property, as the case may be) equivalent to such property.

5.3 **Substitutions of Collateral**

Borrower may from time to time call for the repayment of Cash Collateral or the Delivery of Collateral equivalent to any Collateral delivered to Lender prior to the date on which the same would otherwise have been repayable or deliverable provided that at or prior to the time of such repayment or Delivery Borrower shall have delivered Alternative Collateral acceptable to Lender and Borrower is in compliance with paragraph 5.4 or paragraph 5.5, as applicable.

5.4 Marking to Market of Collateral during the currency of a Loan on aggregated basis

Unless paragraph 1.3 of the Schedule indicates that paragraph 5.5 shall apply in lieu of this paragraph 5.4, or unless otherwise agreed between the Parties:

- the aggregate Market Value of the Collateral delivered to or deposited with Lender (excluding any Equivalent Collateral repaid or delivered under paragraphs 5.4(b) or 5.5(b) (as the case may be)) (*Posted Collateral*) in respect of all Loans outstanding under this Agreement shall equal the aggregate of the Market Value of Securities equivalent to the Loaned Securities and the applicable Margin (the *Required Collateral Value*) in respect of such Loans;
- (b) if at any time on any Business Day the aggregate Market Value of the Posted Collateral in respect of all Loans outstanding under this Agreement together with: (i) all amounts due and payable by the Lender under this Agreement but which are unpaid; and (ii) if agreed between the parties and if the Income Record Date has occurred in respect of any Non-Cash Collateral, the amount or Market Value of Income payable in respect

of such Non-Cash Collateral exceeds the aggregate of the Required Collateral Values in respect of such Loans together with: (i) all amounts due and payable by the Borrower under this Agreement but which are unpaid; and (ii) if agreed between the parties and if the Income Record Date has occurred in respect of any securities equivalent to Loaned Securities, the amount or Market Value of Income payable in respect of such Equivalent Securities, Lender shall (on demand) repay and/or deliver, as the case may be, to Borrower such Equivalent Collateral as will eliminate the excess;

- if at any time on any Business Day the aggregate Market Value of the Posted Collateral in respect of all Loans outstanding under this Agreement together with: (i) all amounts due and payable by the Lender under this Agreement but which are unpaid; and (ii) if agreed between the parties and if the Income Record Date has occurred in respect of any Non-Cash Collateral, the amount or Market Value of Income payable in respect of such Non-Cash Collateral falls below the aggregate of Required Collateral Values in respect of all such Loans together with: (i) all amounts due and payable by the Borrower under this Agreement but which are unpaid; and (ii) if agreed between the parties and if the Income Record Date has occurred in respect of Securities equivalent to any Loaned Securities, the amount or Market Value of Income payable in respect of such Equivalent Securities, Borrower shall (on demand) provide such further Collateral to Lender as will eliminate the deficiency;
- (d) where a Party acts as both Lender and Borrower under this Agreement, the provisions of paragraphs 5.4(b) and 5.4(c) shall apply separately (and without duplication) in respect of Loans entered into by that Party as Lender and Loans entered into by that Party as Borrower.

5.5 Marking to Market of Collateral during the currency of a Loan on a Loan by Loan basis

If paragraph 1.2 of the Schedule indicates this paragraph 5.5 shall apply in lieu of paragraph 5.4, the Posted Collateral in respect of any Loan shall bear from day to day and at any time the same proportion to the Market Value of Securities equivalent to the Loaned Securities as the Posted Collateral bore at the commencement of such Loan. Accordingly:

- (a) the Market Value of the Posted Collateral to be delivered or deposited while the Loan continues shall be equal to the Required Collateral Value;
- (b) if at any time on any Business Day the Market Value of the Posted Collateral in respect of any Loan together with: (i) all amounts due and payable by the Lender in respect of that Loan but which are unpaid; and (ii) if agreed between the parties and if the Income Record Date has occurred in respect of any Non-Cash Collateral, the amount or Market Value of Income payable in respect of such Non-Cash Collateral exceeds the Required Collateral Value in respect of such Loan together with: (i) all amounts due and payable by the Borrower in respect of that Loan; and (ii) if agreed between the parties and if the Income Record Date has occurred in respect of Securities equivalent to any Loaned Securities, the amount or Market Value of Income payable in respect of such Equivalent Securities, Lender shall (on demand) repay and/or deliver, as the case may be, to Borrower such Equivalent Collateral as will eliminate the excess; and
- (c) if at any time on any Business Day the Market Value of the Posted Collateral together with: (i) all amounts due any payable by the Lender in respect of that Loan; and (ii) if agreed between the parties and if the Income Record Date has occurred in respect of any Non-Cash Collateral, the amount or Market Value of Income payable in respect of such Non-Cash Collateral falls below the Required Collateral Value together with: (i) all amounts due and payable by the Borrower in respect of that Loan; and (ii) if agreed between the parties and if the Income Record Date has occurred in respect of

Securities equivalent to any Loaned Securities, the amount or Market Value of Income payable in respect of such Equivalent Securities, Borrower shall (on demand) provide such further Collateral to Lender as will eliminate the deficiency.

5.6 Requirements to deliver excess Collateral

Where paragraph 5.4 applies, unless paragraph 1.4 Schedule of the Schedule indicates that this paragraph 5.6 does not apply, if a Party (the *first Party*) would, but for this paragraph 5.6, be required under paragraph 5.4 to provide further Collateral or deliver Equivalent Collateral in circumstances where the other Party (the *second Party*) would, but for this paragraph 5.6, also be required to or provide Collateral or deliver Equivalent Collateral under paragraph 5.4, then the Market Value of the Collateral or Equivalent Collateral deliverable by the first Party (*X*) shall be set off against the Market Value of the Collateral or Equivalent Collateral deliverable by the second Party (*Y*) and the only obligation of the Parties under paragraph 5.4 shall be, where X exceeds Y, an obligation of the first Party, or where Y exceeds X, an obligation of the second Party to repay and/or (as the case may be) deliver Equivalent Collateral or to deliver further Collateral having a Market Value equal to the difference between X and Y.

5.7 Where Equivalent Collateral is repaid or delivered (as the case may be) or further Collateral is provided by a Party under paragraph 5.6, the Parties shall agree to which Loan or Loans such repayment, delivery or further provision is to be attributed and failing agreement it shall be attributed, as determined by the Party making such repayment, delivery or further provision to the earliest outstanding Loan and, in the case of a repayment or delivery up to the point at which the Market Value of Collateral in respect of such Loan equals the Required Collateral Value in respect of such Loan, and then to the next earliest outstanding Loan up to the similar point and so on.

5.8 Timing of repayments of excess Collateral or deliveries of further Collateral

Where any Equivalent Collateral falls to be repaid or delivered (as the case may be) or further Collateral is to be provided under this paragraph 5, unless otherwise provided or agreed between the Parties, if the relevant demand is received by the Notification Time specified in paragraph 1.5 of the Schedule, then the delivery shall be made not later than the Close of Business on the same Business Day; if a demand is received after the Notification Time, then the relevant delivery shall be made not later than the Close of Business on the next Business Day after the date such demand is received.

5.9 Substitutions and extensions of Letters of Credit

Where Collateral is a Letter of Credit, Lender may by notice to Borrower require that Borrower, on the third Business Day following the date of delivery of such notice (or by such other time as the Parties may agree), substitute Collateral consisting of cash or other Collateral acceptable to Lender for the Letter of Credit. Prior to the expiration of any Letter of Credit supporting Borrower's obligations hereunder, Borrower shall, no later than 10.30 a.m. UK time on the second Business Day prior to the date such Letter of Credit expires (or by such other time as the Parties may agree), obtain an extension of the expiration of such Letter of Credit or replace such Letter of Credit by providing Lender with a substitute Letter of Credit in an amount at least equal to the amount of the Letter of Credit for which it is substituted.

6. DISTRIBUTIONS AND CORPORATE ACTIONS

6.1 In this paragraph 6, references to an amount of Income *received* by any Party in respect of any Loaned Securities or Non-Cash Collateral shall be to an amount received from the issuer after any applicable withholding or deduction for or on account of Tax.

6.2 Manufactured payments in respect of Loaned Securities

Where the term of a Loan extends over an Income Record Date in respect of any Loaned Securities, Borrower shall, on the date such Income is paid by the issuer, or on such other date as the Parties may from time to time agree, pay or deliver to Lender such sum of money or property as is agreed between the Parties or, failing such agreement, a sum of money or property equivalent to (and in the same currency as) the type and amount of such Income that would be received by Lender in respect of such Loaned Securities assuming such Securities were not loaned to Borrower and were retained by Lender on the Income Record Date.

6.3 Manufactured payments in respect of Non-Cash Collateral

Where Non-Cash Collateral is delivered by Borrower to Lender and an Income Record Date in respect of such Non-Cash Collateral occurs before Equivalent Collateral is delivered by Lender to Borrower, Lender shall on the date such Income is paid, or on such other date as the Parties may from time to time agree, pay or deliver to Borrower a sum of money or property as is agreed between the Parties or, failing such agreement, a sum of money or property equivalent to (and in the same currency as) the type and amount of such Income that would be received by Lender in respect of such Non-Cash Collateral assuming Lender:

- (a) retained the Non-Cash Collateral on the Income Record Date; and
- (b) is not entitled to any credit, benefit or other relief in respect of Tax under any Applicable Law.

6.4 Indemnity for failure to redeliver Equivalent Non-Cash Collateral

Unless paragraph 1.6 of the Schedule indicates that this paragraph does not apply, where:

- (a) prior to any Income Record Date in relation to Non-Cash Collateral, Borrower has in accordance with paragraph 5.3 called for the Delivery of Equivalent Non-Cash Collateral:
- (b) Borrower has given notice of such call to Lender so as to be effective, at the latest, five hours before the Close of Business on the last Business Day on which Lender would customarily be required to initiate settlement of the Non-Cash Collateral to enable settlement to take place on the Business Day immediately preceding the relevant Income Record Date;
- (c) Borrower has provided reasonable details to Lender of the Non-Cash Collateral, the relevant Income Record Date and the proposed Alternative Collateral;
- (d) Lender, acting reasonably, has determined that such Alternative Collateral is acceptable to it and Borrower shall have delivered or delivers such Alternative Collateral to Lender; and
- (e) Lender has failed to make reasonable efforts to transfer Equivalent Non-Cash Collateral to Borrower prior to such Income Record Date,

Lender shall indemnify Borrower in respect of any cost, loss or damage (excluding any indirect or consequential loss or damage or any amount otherwise compensated by Lender, including pursuant to paragraphs 6.3 and/or 9.3) suffered by Borrower that it would not have suffered had the relevant Equivalent Non-Cash Collateral been transferred to Borrower prior to such Income Record Date.

6.5 **Income in the form of Securities**

Where Income, in the form of securities, is paid in relation to any Loaned Securities or Collateral, such securities shall be added to such Loaned Securities or Collateral (and shall constitute Loaned Securities or Collateral, as the case may be, and be part of the relevant Loan) and will not be delivered to Lender, in the case of Loaned Securities, or to Borrower, in the case of Collateral, until the end of the relevant Loan, provided that the Lender or Borrower (as the case may be) fulfils its obligations under paragraph 5.4 or 5.5 (as applicable) with respect to the additional Loaned Securities or Collateral, as the case may be.

6.6 Exercise of voting rights

Where any voting rights fall to be exercised in relation to any Loaned Securities or Collateral, neither Borrower, in the case of Equivalent Securities, nor Lender, in the case of Equivalent Collateral, shall have any obligation to arrange for voting rights of that kind to be exercised in accordance with the instructions of the other Party in relation to the Securities borrowed by it or transferred to it by way of Collateral, as the case may be, unless otherwise agreed between the Parties.

6.7 **Corporate actions**

Where, in respect of any Loaned Securities or any Collateral, any rights relating to conversion, sub-division, consolidation, pre-emption, rights arising under a takeover offer, rights to receive securities or a certificate which may at a future date be exchanged for securities or other rights, including those requiring election by the holder for the time being of such Securities or Collateral, become exercisable prior to the delivery of Equivalent Securities or Equivalent Collateral, then Lender or Borrower, as the case may be, may, within a reasonable time before the latest time for the exercise of the right or option give written notice to the other Party that on delivery of Equivalent Securities or Equivalent Collateral, as the case may be, it wishes to receive Equivalent Securities or Equivalent Collateral in such form as will arise if the right is exercised or, in the case of a right which may be exercised in more than one manner, is exercised as is specified in such written notice.

7. RATES APPLICABLE TO LOANED SECURITIES AND CASH COLLATERAL

7.1 Rates in respect of Loaned Securities

In respect of each Loan, Borrower shall pay to Lender, in the manner prescribed in sub-paragraph 7.3, sums calculated by applying such rate as shall be agreed between the Parties from time to time to the daily Market Value of the Loaned Securities.

7.2 Rates in respect of Cash Collateral

Where Cash Collateral is deposited with Lender in respect of any Loan, Lender shall pay to Borrower, in the manner prescribed in paragraph 7.3, sums calculated by applying such rates as shall be agreed between the Parties from time to time to the amount of such Cash Collateral. Any such payment due to Borrower may be set-off against any payment due to Lender pursuant to paragraph 7.1.

7.3 **Payment of rates**

In respect of each Loan, the payments referred to in paragraph 7.1 and 7.2 shall accrue daily in respect of the period commencing on and inclusive of the Settlement Date and terminating on and exclusive of the Business Day upon which Equivalent Securities are delivered or Cash Collateral is repaid. Unless otherwise agreed, the sums so accruing in respect of each calendar month shall be paid in arrears by the relevant Party not later than the Business Day which is the tenth Business Day after the last Business Day of the calendar month to which such payments relate or such other date as the Parties shall from time to time agree.

8. DELIVERY OF EQUIVALENT SECURITIES

8.1 Lender's right to terminate a Loan

Subject to paragraph 11 and the terms of the relevant Loan, Lender shall be entitled to terminate a Loan and to call for the delivery of all or any Equivalent Securities at any time by giving notice on any Business Day of not less than the standard settlement time for such Equivalent Securities on the exchange or in the clearing organisation through which the Loaned Securities were originally delivered. Borrower shall deliver such Equivalent Securities not later than the expiry of such notice in accordance with Lender's instructions.

8.2 **Borrower's right to terminate a Loan**

Subject to the terms of the relevant Loan, Borrower shall be entitled at any time to terminate a Loan and to deliver all and any Equivalent Securities due and outstanding to Lender in accordance with Lender's instructions and Lender shall accept such delivery.

8.3 Delivery of Equivalent Securities on termination of a Loan

Borrower shall procure the Delivery of Equivalent Securities to Lender or deliver Equivalent Securities in accordance with this Agreement and the terms of the relevant Loan on termination of the Loan. For the avoidance of doubt any reference in this Agreement or in any other agreement or communication between the Parties (howsoever expressed) to an obligation to deliver or account for or act in relation to Loaned Securities shall accordingly be construed as a reference to an obligation to deliver or account for or act in relation to Equivalent Securities.

8.4 Delivery of Equivalent Collateral on termination of a Loan

On the date and time that Equivalent Securities are required to be delivered by Borrower on the termination of a Loan, Lender shall simultaneously (subject to paragraph 5.4 if applicable) repay to Borrower any Cash Collateral or, as the case may be, deliver Collateral equivalent to the Collateral provided by Borrower pursuant to paragraph 5 in respect of such Loan. For the avoidance of doubt any reference in this Agreement or in any other agreement or communication between the Parties (however expressed) to an obligation to deliver or account for or act in relation to Collateral shall accordingly be construed as a reference to an obligation to deliver or account for or act in relation to Equivalent Collateral.

8.5 **Delivery of Letters of Credit**

Where a Letter of Credit is provided by way of Collateral, the obligation to deliver Equivalent Collateral is satisfied by Lender delivering for cancellation the Letter of Credit so provided, or where the Letter of Credit is provided in respect of more than one Loan, by Lender consenting to a reduction in the value of the Letter of Credit.

8.6 **Delivery obligations to be reciprocal**

Neither Party shall be obliged to make delivery (or make a payment as the case may be) to the other unless it is satisfied that the other Party will make such delivery (or make an appropriate payment as the case may be) to it. If it is not so satisfied (whether because an Event of Default has occurred in respect of the other Party or otherwise) it shall notify the other Party and unless that other Party has made arrangements which are sufficient to assure full delivery (or the appropriate payment as the case may be) to the notifying Party, the notifying Party shall (provided it is itself in a position, and willing, to perform its own obligations) be entitled to withhold delivery (or payment, as the case may be) to the other Party until such arrangements to assure full delivery (or the appropriate payment as the case may be) are made.

9. FAILURE TO DELIVER

9.1 Borrower's failure to deliver Equivalent Securities

If Borrower fails to deliver Equivalent Securities in accordance with paragraph 8.3 Lender may:

- (a) elect to continue the Loan (which, for the avoidance of doubt, shall continue to be taken into account for the purposes of paragraph 5.4 or 5.5 as applicable); or
- (b) at any time while such failure continues, by written notice to Borrower declare that that Loan (but only that Loan) shall be terminated immediately in accordance with paragraph 11.2 as if (i) an Event of Default had occurred in relation to the Borrower, (ii) references to the Termination Date were to the date on which notice was given under this sub-paragraph, and (iii) the Loan were the only Loan outstanding. For the avoidance of doubt, any such failure shall not constitute an Event of Default (including under paragraph 10(i)) unless the Parties otherwise agree.

9.2 Lender's failure to deliver Equivalent Collateral

If Lender fails to deliver Equivalent Collateral comprising Non-Cash Collateral in accordance with paragraph 8.4 or 8.5, Borrower may:

- (a) elect to continue the Loan (which, for the avoidance of doubt, shall continue to be taken into account for the purposes of paragraph 5.4 or 5.5 as applicable); or
- (b) at any time while such failure continues, by written notice to Lender declare that that Loan (but only that Loan) shall be terminated immediately in accordance with paragraph 11.2 as if (i) an Event of Default had occurred in relation to the Lender, (ii) references to the Termination Date were to the date on which notice was given under this sub-paragraph, and (iii) the Loan were the only Loan outstanding. For the avoidance of doubt, any such failure shall not constitute an Event of Default (including under paragraph 10.1(i)) unless the Parties otherwise agree.

9.3 Failure by either Party to deliver

Where a Party (the *Transferor*) fails to deliver Equivalent Securities or Equivalent Collateral by the time required under this Agreement or within such other period as may be agreed between the Transferor and the other Party (the *Transferee*) and the Transferee:

(a) incurs interest, overdraft or similar costs and expenses; or

(b) incurs costs and expenses as a direct result of a Buy-in exercised against it by a third party,

then the Transferor agrees to pay within one Business Day of a demand from the Transferee and hold harmless the Transferee with respect to all reasonable costs and expenses listed in sub-paragraphs (a) and (b) 如上 above properly incurred which arise directly from such failure other than (i) such costs and expenses which arise from the negligence or wilful default of the Transferee and (ii) any indirect or consequential losses.

10. EVENTS OF DEFAULT

- 10.1 Each of the following events occurring and continuing in relation to either Party (the *Defaulting Party*), the other Party being the *Non-Defaulting Party*) shall be an Event of Default but only (subject to sub-paragraph10.1(d)) where the Non-Defaulting Party serves written notice on the Defaulting Party:
 - (a) Borrower or Lender failing to pay or repay Cash Collateral or to deliver Collateral on commencement of the Loan under paragraph 5.1 or to deliver further Collateral under paragraph 5.4 or 5.5;
 - (b) Lender or Borrower failing to comply with its obligations under paragraph 6.2 or 6.3 upon the due date and not remedying such failure within three Business Days after the Non-Defaulting Party serves written notice requiring it to remedy such failure;
 - (c) Lender or Borrower failing to pay any sum due under paragraph 9.1(b), 9.2(b) or 9.3 upon the due date;
 - (d) an Act of Insolvency occurring with respect to Lender or Borrower, provided that, where the Parties have specified in paragraph 5 of the Schedule that Automatic Early Termination shall apply, an Act of Insolvency which is the presentation of a petition for winding up or any analogous proceeding or the appointment of a liquidator or analogous officer of the Defaulting Party shall not require the Non-Defaulting Party to serve written notice on the Defaulting Party (*Automatic Early Termination*);
 - (e) any warranty made by Lender or Borrower in paragraph 13 or paragraphs 14(a) to 14(d) being incorrect or untrue in any material respect when made or repeated or deemed to have been made or repeated;
 - (f) Lender or Borrower admitting to the other that it is unable to, or it intends not to, perform any of its obligations under this Agreement and/or in respect of any Loan where such failure to perform would with the service of notice or lapse of time constitute an Event of Default;
 - (g) all or any material part of the assets of Lender or Borrower being transferred or ordered to be transferred to a trustee (or a person exercising similar functions) by a regulatory authority pursuant to any legislation;
 - (h) Lender (if applicable) or Borrower being declared in default or being suspended or expelled from membership of or participation in, any securities exchange or suspended or prohibited from dealing in securities by any regulatory authority, in each case on the grounds that it has failed to meet any requirements relating to financial resources or credit rating; or

- (i) Lender or Borrower failing to perform any other of its obligations under this Agreement and not remedying such failure within 30 days after the Non-Defaulting Party serves written notice requiring it to remedy such failure.
- 10.2 Each Party shall notify the other (in writing) if an Event of Default or an event which, with the passage of time and/or upon the serving of a written notice as referred to above, would be an Event of Default, occurs in relation to it.
- 10.3 The provisions of this Agreement constitute a complete statement of the remedies available to each Party in respect of any Event of Default.
- 10.4 Subject to paragraphs 9 and 11, neither Party may claim any sum by way of consequential loss or damage in the event of failure by the other Party to perform any of its obligations under this Agreement.

11. CONSEQUENCES OF AN EVENT OF DEFAULT

- 11.1 If an Event of Default occurs in relation to either Party then paragraphs 11.2 to 11.7 below shall apply.
- 11.2 The Parties' delivery and payment obligations (and any other obligations they have under this Agreement) shall be accelerated so as to require performance thereof at the time such Event of Default occurs (the date of which shall be the *Termination Date*) so that performance of such delivery and payment obligations shall be effected only in accordance with the following provisions.
 - (a) The Default Market Value of the Equivalent Securities and Equivalent Non-Cash Collateral to be delivered and the amount of any Cash Collateral (including sums accrued) to be repaid and any other cash (including interest accrued) to be paid by each Party shall be established by the Non-Defaulting Party in accordance with paragraph 11.4 and deemed as at the Termination Date.
 - (b) On the basis of the sums so established, an account shall be taken (as at the Termination Date) of what is due from each Party to the other under this Agreement (on the basis that each Party's claim against the other in respect of delivery of Equivalent Securities or Equivalent Non-Cash Collateral equal to the Default Market Value thereof) and the sums due from one Party shall be set off against the sums due from the other and only the balance of the account shall be payable (by the Party having the claim valued at the lower amount pursuant to the foregoing) and such balance shall be payable on the next following Business Day after such account has been taken and such sums have been set off in accordance with this paragraph. For the purposes of this calculation, any sum not denominated in the Base Currency shall be converted into the Base Currency at the Spot Rate prevailing at such dates and times determined by the Non-Defaulting Party acting reasonably.
 - (c) If the balance under sub-paragraph (b) above is payable by the Non-Defaulting Party and the Non-Defaulting Party had delivered to the Defaulting Party a Letter of Credit, the Defaulting Party shall draw on the Letter of Credit to the extent of the balance due and shall subsequently deliver for cancellation the Letter of Credit so provided.
 - (d) If the balance under sub-paragraph (b) above is payable by the Defaulting Party and the Defaulting Party had delivered to the Non-Defaulting Party a Letter of Credit, the Non-Defaulting Party shall draw on the Letter of Credit to the extent of the balance due and shall subsequently deliver for cancellation the Letter of Credit so provided.

- (e) In all other circumstances, where a Letter of Credit has been provided to a Party, such Party shall deliver for cancellation the Letter of Credit so provided.
- 11.3 For the purposes of this Agreement, the *Default Market Value* of any Equivalent Collateral in the form of a Letter of Credit shall be zero and of any Equivalent Securities or any other Equivalent Non-Cash Collateral shall be determined in accordance with paragraphs 11.4 to 11.6 below, and for this purpose:
 - (a) the *Appropriate Market* means, in relation to securities of any description, the market which is the most appropriate market for securities of that description, as determined by the Non-Defaulting Party;
 - (b) the *Default Valuation Time* means, in relation to an Event of Default, the close of business in the Appropriate Market on the fifth dealing day after the day on which that Event of Default occurs or, where that Event of Default is the occurrence of an Act of Insolvency in respect of which under paragraph 10.1(d) no notice is required from the Non-Defaulting Party in order for such event to constitute an Event of Default, the close of business on the fifth dealing day after the day on which the Non-Defaulting Party first became aware of the occurrence of such Event of Default;
 - (c) **Deliverable Securities** means Equivalent Securities or Equivalent Non-Cash Collateral to be delivered by the Defaulting Party;
 - (d) Net Value means at any time, in relation to any Deliverable Securities or Receivable Securities, the amount which, in the reasonable opinion of the Non-Defaulting Party, represents their fair market value, having regard to such pricing sources and methods (which may include, without limitation, available prices for securities with similar maturities, terms and credit characteristics as the relevant Equivalent Securities or Equivalent Collateral) as the Non-Defaulting Party considers appropriate, less, in the case of Receivable Securities, or plus, in the case of Deliverable Securities, all Transaction Costs incurred or reasonably anticipated in connection with the purchase or sale of such securities;
 - (e) **Receivable Securities** means Equivalent Securities or Equivalent Non-Cash Collateral to be delivered to the Defaulting Party; and
 - (f) *Transaction Costs* in relation to any transaction contemplated in paragraph 11.4 or 11.5 means the reasonable costs, commissions (including internal commissions), fees and expenses (including any mark-up or mark-down or premium paid for guaranteed delivery) incurred or reasonably anticipated in connection with the purchase of Deliverable Securities or sale of Receivable Securities, calculated on the assumption that the aggregate thereof is the least that could reasonably be expected to be paid in order to carry out the transaction.
- 11.4 If between the Termination Date and the Default Valuation Time:
 - (a) the Non-Defaulting Party has sold, in the case of Receivable Securities, or purchased, in the case of Deliverable Securities, securities which form part of the same issue and are of an identical type and description as those Equivalent Securities or that Equivalent Collateral, (and regardless as to whether or not such sales or purchases have settled) the Non-Defaulting Party may elect to treat as the Default Market Value:
 - (i) in the case of Receivable Securities, the net proceeds of such sale after deducting all Transaction Costs; provided that, where the securities sold are not identical in amount to the Equivalent Securities or Equivalent Collateral,

the Non-Defaulting Party may, acting in good faith, either (A) elect to treat such net proceeds of sale divided by the amount of securities sold and multiplied by the amount of the Equivalent Securities or Equivalent Collateral as the Default Market Value or (B) elect to treat such net proceeds of sale of the Equivalent Securities or Equivalent Collateral actually sold as the Default Market Value of that proportion of the Equivalent Securities or Equivalent Collateral, and, in the case of (B), the Default Market Value of the balance of the Equivalent Securities or Equivalent Collateral shall be determined separately in accordance with the provisions of this paragraph 11.4; or

- in the case of Deliverable Securities, the aggregate cost of such purchase, including all Transaction Costs; provided that, where the securities purchased are not identical in amount to the Equivalent Securities or Equivalent Collateral, the Non-Defaulting Party may, acting in good faith, either (A) elect to treat such aggregate cost divided by the amount of securities purchased and multiplied by the amount of the Equivalent Securities or Equivalent Collateral as the Default Market Value or (B) elect to treat the aggregate cost of purchasing the Equivalent Securities or Equivalent Collateral actually purchased as the Default Market Value of that proportion of the Equivalent Securities or Equivalent Collateral, and, in the case of (B), the Default Market Value of the balance of the Equivalent Securities or Equivalent Collateral shall be determined separately in accordance with the provisions of this paragraph 11.4;
- (b) the Non-Defaulting Party has received, in the case of Deliverable Securities, offer quotations or, in the case of Receivable Securities, bid quotations in respect of securities of the relevant description from two or more market makers or regular dealers in the Appropriate Market in a commercially reasonable size (as determined by the Non-Defaulting Party) the Non-Defaulting Party may elect to treat as the Default Market Value of the relevant Equivalent Securities or Equivalent Collateral:
 - (i) the price quoted (or where more than one price is so quoted, the arithmetic mean of the prices so quoted) by each of them for, in the case of Deliverable Securities, the sale by the relevant market marker or dealer of such securities or, in the case of Receivable Securities, the purchase by the relevant market maker or dealer of such securities, provided that such price or prices quoted may be adjusted in a commercially reasonable manner by the Non-Defaulting Party to reflect accrued but unpaid coupons not reflected in the price or prices quoted in respect of such Securities;
 - (ii) after deducting, in the case of Receivable Securities or adding in the case of Deliverable Securities the Transaction Costs which would be incurred or reasonably anticipated in connection with such transaction.
- If, acting in good faith, either (A) the Non-Defaulting Party has endeavoured but been unable to sell or purchase securities in accordance with paragraph 11.4(a) above or to obtain quotations in accordance with paragraph 11.4(b) above (or both) or (B) the Non-Defaulting Party has determined that it would not be commercially reasonable to sell or purchase securities at the prices bid or offered or to obtain such quotations, or that it would not be commercially reasonable to use any quotations which it has obtained under paragraph 11.4(b) above the Non-Defaulting Party may determine the Net Value of the relevant Equivalent Securities or Equivalent Collateral (which shall be specified) and the Non-Defaulting Party may elect to treat such Net Value as the Default Market Value of the relevant Equivalent Securities or Equivalent Collateral.

11.6 To the extent that the Non-Defaulting Party has not determined the Default Market Value in accordance with paragraph 11.4, the Default Market Value of the relevant Equivalent Securities or Equivalent Collateral shall be an amount equal to their Net Value at the Default Valuation Time; provided that, if at the Default Valuation Time the Non-Defaulting Party reasonably determines that, owing to circumstances affecting the market in the Equivalent Securities or Equivalent Collateral in question, it is not reasonably practicable for the Non-Defaulting Party to determine a Net Value of such Equivalent Securities or Equivalent Collateral which is commercially reasonable (by reason of lack of tradable prices or otherwise), the Default Market Value of such Equivalent Securities or Equivalent Collateral shall be an amount equal to their Net Value as determined by the Non-Defaulting Party as soon as reasonably practicable after the Default Valuation Time.

Other costs, expenses and interest payable in consequence of an Event of Default

11.7 The Defaulting Party shall be liable to the Non-Defaulting Party for the amount of all reasonable legal and other professional expenses incurred by the Non-Defaulting Party in connection with or as a consequence of an Event of Default, together with interest thereon at such rate as is agreed by the Parties and specified in paragraph 10 of the Schedule or, failing such agreement, the overnight London Inter Bank Offered Rate as quoted on a reputable financial information service (*LIBOR*) as at 11.00 a.m., London time, on the date on which it is to be determined or, in the case of an expense attributable to a particular transaction and, where the Parties have previously agreed a rate of interest for the transaction, that rate of interest if it is greater than LIBOR. Interest will accrue daily on a compound basis.

Set-off

11.8 Any amount payable to one Party (the *Payee*) by the other Party (the *Payer*) under paragraph 11.2(b) may, at the option of the Non-Defaulting Party, be reduced by its set-off against any amount payable (whether at such time or in the future or upon the occurrence of a contingency) by the Payee to the Payer (irrespective of the currency, place of payment or booking office of the obligation) under any other agreement between the Payee and the Payer or instrument or undertaking issued or executed by one Party to, or in favour of, the other Party. If an obligation is unascertained, the Non-Defaulting Party may in good faith estimate that obligation and set off in respect of the estimate, subject to accounting to the other Party when the obligation is ascertained. Nothing in this paragraph shall be effective to create a charge or other security interest. This paragraph shall be without prejudice and in addition to any right of set-off, combination of accounts, lien or other right to which any Party is at any time otherwise entitled (whether by operation of law, contract or otherwise).

12. TAXES

Withholding, gross-up and provision of information

- 12.1 All payments under this Agreement shall be made without any deduction or withholding for or on account of any Tax unless such deduction or withholding is required by any Applicable Law.
- 12.2 Except as otherwise agreed, if the paying Party is so required to deduct or withhold, then that Party (*Payer*) shall:
 - (a) promptly notify the other Party (*Recipient*) of such requirement;
 - (b) pay or otherwise account for the full amount required to be deducted or withheld to the relevant authority;

- (c) upon written demand of Recipient, forward to Recipient documentation reasonably acceptable to Recipient, evidencing such payment to such authorities; and
- (d) other than in respect of any payment made by Lender to Borrower under paragraph 6.3, pay to Recipient, in addition to the payment to which Recipient is otherwise entitled under this Agreement, such additional amount as is necessary to ensure that the amount actually received by Recipient (after taking account of such withholding or deduction) will equal the amount Recipient would have received had no such deduction or withholding been required; provided Payer will not be required to pay any additional amount to Recipient under this sub-paragraph (d) to the extent it would not be required to be paid but for the failure by Recipient to comply with or perform any obligation under paragraph 12.3.
- 12.3 Each Party agrees that it will upon written demand of the other Party deliver to such other Party (or to any government or other taxing authority as such other Party directs), any form or document and provide such other cooperation or assistance as may (in either case) reasonably be required in order to allow such other Party to make a payment under this Agreement without any deduction or withholding for or on account of any Tax or with such deduction or withholding at a reduced rate (so long as the completion, execution or submission of such form or document, or the provision of such cooperation or assistance, would not materially prejudice the legal or commercial position of the Party in receipt of such demand). Any such form or document shall be accurate and completed in a manner reasonably satisfactory to such other Party and shall be executed and delivered with any reasonably required certification by such date as is agreed between the Parties or, failing such agreement, as soon as reasonably practicable.

Stamp Tax

- 12.4 Unless otherwise agreed, Borrower hereby undertakes promptly to pay and account for any Stamp Tax chargeable in connection with any transaction effected pursuant to or contemplated by this Agreement (other than any Stamp Tax that would not be chargeable but for Lender's failure to comply with its obligations under this Agreement).
- Borrower shall indemnify and keep indemnified Lender against any liability arising as a result of Borrower's failure to comply with its obligations under paragraph 12.4.

Sales Tax

12.6 All sums payable by one Party to another under this Agreement are exclusive of any Sales Tax chargeable on any supply to which such sums relate and an amount equal to such Sales Tax shall in each case be paid by the Party making such payment on receipt of an appropriate Sales Tax invoice.

Retrospective changes in law

- 12.7 Unless otherwise agreed, amounts payable by one Party to another under this Agreement shall be determined by reference to Applicable Law as at the date of the relevant payment and no adjustment shall be made to amounts paid under this Agreement as a result of:
 - (a) any retrospective change in Applicable Law which is announced or enacted after the date of the relevant payment; or
 - (b) any decision of a court of competent jurisdiction which is made after the date of the relevant payment (other than where such decision results from an action taken with respect to this Agreement or amounts paid or payable under this Agreement).

13. LENDER'S WARRANTIES

Each Party hereby warrants and undertakes to the other on a continuing basis to the intent that such warranties shall survive the completion of any transaction contemplated herein that, where acting as a Lender:

- (a) it is duly authorised and empowered to perform its duties and obligations under this Agreement;
- (b) it is not restricted under the terms of its constitution or in any other manner from lending Securities in accordance with this Agreement or from otherwise performing its obligations hereunder;
- (c) it is absolutely entitled to pass full legal and beneficial ownership of all Securities provided by it hereunder to Borrower free from all liens, charges and encumbrances; and
- (d) it is acting as principal in respect of this Agreement, other than in respect of an Agency

14. BORROWER'S WARRANTIES

Each Party hereby warrants and undertakes to the other on a continuing basis to the intent that such warranties shall survive the completion of any transaction contemplated herein that, where acting as a Borrower:

- (a) it has all necessary licences and approvals, and is duly authorised and empowered, to perform its duties and obligations under this Agreement and will do nothing prejudicial to the continuation of such authorisation, licences or approvals;
- (b) it is not restricted under the terms of its constitution or in any other manner from borrowing Securities in accordance with this Agreement or from otherwise performing its obligations hereunder;
- (c) it is absolutely entitled to pass full legal and beneficial ownership of all Collateral provided by it hereunder to Lender free from all liens, charges and encumbrances;
- (d) it is acting as principal in respect of this Agreement; and
- (e) it is not entering into a Loan for the primary purpose of obtaining or exercising voting rights in respect of the Loaned Securities.

15. INTEREST ON OUTSTANDING PAYMENTS

In the event of either Party failing to remit sums in accordance with this Agreement such Party hereby undertakes to pay to the other Party upon demand interest (before as well as after judgment) on the net balance due and outstanding, for the period commencing on and inclusive of the original due date for payment to (but excluding) the date of actual payment, in the same currency as the principal sum and at the rate referred to in paragraph 11.7. Interest will accrue daily on a compound basis and will be calculated according to the actual number of days elapsed. No interest shall be payable under this paragraph in respect of any day on which one Party endeavours to make a payment to the other Party but the other Party is unable to receive it.

16. TERMINATION OF THIS AGREEMENT

Each Party shall have the right to terminate this Agreement by giving not less than 15 Business Days' notice in writing to the other Party (which notice shall specify the date of termination) subject to an obligation to ensure that all Loans which have been entered into but not discharged at the time such notice is given are duly discharged in accordance with this Agreement.

17. SINGLE AGREEMENT

Each Party acknowledges that, and has entered into this Agreement and will enter into each Loan in consideration of and in reliance upon the fact that, all Loans constitute a single business and contractual relationship and are made in consideration of each other. Accordingly, each Party agrees:

- (a) to perform all of its obligations in respect of each Loan, and that a default in the performance of any such obligations shall constitute a default by it in respect of all Loans, subject always to the other provisions of the Agreement; and
- (b) that payments, deliveries and other transfers made by either of them in respect of any Loan shall be deemed to have been made in consideration of payments, deliveries and other transfers in respect of any other Loan.

18. SEVERANCE

If any provision of this Agreement is declared by any judicial or other competent authority to be void or otherwise unenforceable, that provision shall be severed from the Agreement and the remaining provisions of this Agreement shall remain in full force and effect. The Agreement shall, however, thereafter be amended by the Parties in such reasonable manner so as to achieve as far as possible, without illegality, the intention of the Parties with respect to that severed provision.

19. SPECIFIC PERFORMANCE

Each Party agrees that in relation to legal proceedings it will not seek specific performance of the other Party's obligation to deliver Securities, Equivalent Securities, Collateral or Equivalent Collateral but without prejudice to any other rights it may have.

20. NOTICES

- Any notice or other communication in respect of this Agreement may be given in any manner set forth below to the address or number or in accordance with the electronic messaging system details set out in paragraph 5 of the Schedule and will be deemed effective as indicated:
 - (a) if in writing and delivered in person or by courier, on the date it is delivered;
 - (b) if sent by facsimile transmission, on the date that transmission is received by a responsible employee of the recipient in legible form (it being agreed that the burden of proving receipt will be on the sender and will not be met by a transmission report generated by the sender's facsimile machine);
 - (c) if sent by certified or registered mail (airmail, if overseas) or the equivalent (return receipt requested), on the date that mail is delivered or its delivery is attempted; or
 - (d) if sent by electronic messaging system, on the date that electronic message is received,

unless the date of that delivery (or attempted delivery) or the receipt, as applicable, is not a Business Day or that communication is delivered (or attempted) or received, as applicable, after the Close of Business on a Business Day, in which case that communication shall be deemed given and effective on the first following day that is a Business Day.

20.2 Either Party may by notice to the other change the address or facsimile number or electronic messaging system details at which notices or other communications are to be given to it.

21. ASSIGNMENT

- 21.1 Subject to paragraph 21.2, neither Party may charge, assign or otherwise deal with all or any of its rights or obligations hereunder without the prior consent of the other Party.
- 21.2 Paragraph 21.1 shall not preclude a party from charging, assigning or otherwise dealing with all or any part of its interest in any sum payable to it under paragraph 11.2(b) or 11.7.

22. Non-Waiver

No failure or delay by either Party (whether by course of conduct or otherwise) to exercise any right, power or privilege hereunder shall operate as a waiver thereof nor shall any single or partial exercise of any right, power or privilege preclude any other or further exercise thereof or the exercise of any other right, power or privilege as herein provided.

23. GOVERNING LAW AND JURISDICTION

- 23.1 This Agreement and any non-contractual obligations arising out of or in connection with this Agreement shall be governed by, and shall be construed in accordance with, English law.
- 23.2 The courts of England have exclusive jurisdiction to hear and decide any suit, action or proceedings, and to settle any disputes or any non-contractual obligation which may arise out of or in connection with this Agreement (respectively, *Proceedings* and *Disputes*) and, for these purposes, each Party irrevocably submits to the jurisdiction of the courts of England.
- 23.3 Each Party irrevocably waives any objection which it might at any time have to the courts of England being nominated as the forum to hear and decide any Proceedings and to settle any Disputes and agrees not to claim that the courts of England are not a convenient or appropriate forum.
- 23.4 Each Party hereby respectively appoints the person identified in paragraph 7 of the Schedule pertaining to the relevant Party as its agent to receive on its behalf service of process in the courts of England. If such an agent ceases to be an agent of a Party, the relevant Party shall promptly appoint, and notify the other Party of the identity of its new agent in England.

24. TIME

Time shall be of the essence of the Agreement.

25. RECORDING

The Parties agree that each may record all telephone conversations between them.

26. WAIVER OF IMMUNITY

Each Party hereby waives all immunity (whether on the basis of sovereignty or otherwise) from jurisdiction, attachment (both before and after judgement) and execution to which it might otherwise be entitled in any action or proceeding in the courts of England or of any

other country or jurisdiction relating in any way to this Agreement and agrees that it will not raise, claim or cause to be pleaded any such immunity at or in respect of any such action or proceeding.

27. MISCELLANEOUS

- 27.1 This Agreement constitutes the entire agreement and understanding of the Parties with respect to its subject matter and supersedes all oral communication and prior writings with respect thereto.
- 27.2 The Party (the *Relevant Party*) who has prepared the text of this Agreement for execution (as indicated in paragraph 9 of the Schedule) warrants and undertakes to the other Party that such text conforms exactly to the text of the standard form Global Master Securities Lending Agreement (2010 version) posted by the International Securities Lending Association on its website except as notified by the Relevant Party to the other Party in writing prior to the execution of this Agreement.
- 27.3 Unless otherwise provided for in this Agreement, no amendment in respect of this Agreement will be effective unless in writing (including a writing evidenced by a facsimile transmission) and executed by each of the Parties or confirmed by an exchange of telexes or electronic messages on an electronic messaging system.
- 27.4 The Parties agree that where paragraph 11 of the Schedule indicates that this paragraph 27.4 applies, this Agreement shall apply to all loans which are outstanding as at the date of this Agreement and which are subject to the securities lending agreement or agreements specified in paragraph 11 of the Schedule, and such Loans shall be treated as if they had been entered into under this Agreement, and the terms of such loans are amended accordingly with effect from the date of this Agreement.
- 27.5 The Parties agree that where paragraph 12 of the Schedule indicates that this paragraph 27.5 applies, each may use the services of a third party vendor to automate the processing of Loans under this Agreement and that any data relating to such Loans received from the other Party may be disclosed to such third party vendors.
- 27.6 The obligations of the Parties under this Agreement will survive the termination of any Loan.
- 27.7 The warranties contained in paragraphs 13, 14 and 27.2 and in the Agency Annex will survive termination of this Agreement for so long as any obligations of either of the Parties pursuant to this Agreement remain outstanding.
- 27.8 Except as provided in this Agreement, the rights, powers, remedies and privileges provided in this Agreement are cumulative and not exclusive of any rights, powers, remedies and privileges provided by law.
- 27.9 This Agreement (and each amendment in respect of it) may be executed and delivered in counterparts (including by facsimile transmission), each of which will be deemed an original.
- 27.10 A person who is not a party to this Agreement has no right under the Contracts (Rights of Third Parties) Act 1999 to enforce any terms of this Agreement, but this does not affect any right or remedy of a third party which exists or is available apart from that Act.

EXECUTED by the **PARTIES**

SIGNED by)
)
)
)
)

duly authorised for and on behalf of Guotai Junan Securities (Hong Kong) Limited I have carefully read and understood this Agreement, including the Risk Disclosure Statement Annex, and that the contents of this Agreement, including the Risk Disclosure Statement Annex, was provided in a language of my choice. I accept all the terms of this Agreement and agree to be bound by it as amended and supplemented from time to time.

I acknowledge that I have been invited to read and ask questions about this Agreement, including the Risk Disclosure Statement Annex and I am invited to take independent advice if I wish. I am satisfied with the explanation and clarification provided to me. I declare that I am willing and capable of fulfilling the duties and obligations as specified in this Agreement.

SIGNED by		
holder of		
number		
Representative who has provided and explained to Party B the content of this Agreement including the Risk Disclosure Statement Annex in the language of Party B's choice; and has invited Party B to read the Risk Disclosure Statement Annex, to ask questions and to take independent advice if Party B wish.		
Signature of Licensed Representative		
Full name of Licensed Representative		
CE Number of Licensed Representative		
Date		

SCHEDULE

1. COLLATERAL

- 1.1 The securities, financial instruments and deposits of currency set out in the table below with a cross marked next to them are acceptable forms of Collateral under this Agreement. The parties may, by mutual agreement, amend or vary the acceptable forms of Collateral under this Agreement.
- 1.2 Unless otherwise agreed between the Parties, the Market Value of the Collateral delivered pursuant to paragraph 5 by Borrower to Lender under the terms and conditions of this Agreement shall on each Business Day represent not less than the Market Value of the Loaned Securities together with the percentage contained in the row of the table below corresponding to the particular form of Collateral, referred to in this Agreement as the *Margin*.

Security/Financial	Mark "X" if	Margin
Instrument/Deposit of Currency	acceptable form of Collateral	(%)
Hong Kong dollars cash	X	(i) in respect of a Loan where Party A is the Borrower, 5%; OR (ii) in respect of a Loan where Party B is the Borrower, 20%.

1.3 Basis of Margin Maintenance:

Paragraph 5.4 (aggregation) shall apply.

- 1.4 Paragraph 5.6 (netting of obligations to deliver Collateral and redeliver Equivalent Collateral) shall apply.
- 1.5 For the purposes of Paragraph 5.8, Notification Time means 1:00 p.m, Hong Kong time.
- 1.6 Paragraph 6.4 (indemnity for failure to redeliver Equivalent Non-Cash Collateral) shall apply.

2. BASE CURRENCY

The Base Currency applicable to this Agreement is Hong Kong Dollars

3. PLACES OF BUSINESS

Hong Kong

(See definition of Business Day.)

4. MARKET VALUE

(See definition of Market Value)

5. EVENTS OF DEFAULT

Automatic Early Termination shall not apply in respect of Party A

Automatic Early Termination shall not apply in respect of Party B

6. DESIGNATED OFFICE AND ADDRESS FOR NOTICES

Designated office of Party A:

(a)

	(i)	Address for notices or communic	cations to Party A:
		Address:	27/F, Low Block, Grand Millennium Plaza, 181 Queen's Road Central, Hong Kong
		Attention:	Operations Department
		Facsimile No:	+852 2509 9559
		Telephone No:	+852 2509 7556
		Email::	otc.ops@gtjas.com.hk; collateral.risk@gtjas.com.hk
			mo.tradesupport@gtjas.com.hk
			mo.productcontrol@gtjas.com.hk
			sbl.ops@gtjas.com.hk
	(ii)		(Events of Default) of this Agreement, a copy of e sent to the following address in addition to the
		Address:	27/F, Low Block, Grand Millennium Plaza, 181 Queen's Road Central, Hong Kong
		Attention:	Legal Department
		Telephone:	+852 2509 9118
		Facsimile No:	+852 2185 7130
		Email:	legal@gtjas.com.hk
(b)	Designate	ed office of Party B:	
	Address f	for notices or communications to Pa	arty B:
	Address:		
	Attention	n:	
	Facsimil	e No:	
	Telephoi	ne No:	
	-		

	Electronic Messagi	ng System Details:	·	
7. (a)	Agent of Party A fo	or Service of Proces	SS .	
	Name: Not Applicable			
	Address: Not Appl	icable		
(b)	Agent of Party B fo	or Service of Proces	s	
	Name:	Not Applicable	Please fill in	
	Address:	Not Applicable	Please fill in	

8. AGENCY

- Party A will not act as agent
- Party B will not act as agent
- The Addendum for Pooled Principal Transactions may not apply to Party A
- The Addendum for Pooled Principal Transactions may not apply to Party B

9. PARTY PREPARING THIS AGREEMENT

Party A

10. **DEFAULT INTEREST**

Rate of default interest: Prime rate plus 5% per annum

For the purposes of the foregoing, "**Prime Rate**" means the highest prime rate as quoted on a reputable financial information service as at 11.00 a.m., Hong Kong time on the date on which it is to be determined.

11. EXISTING LOANS

Paragraph 27.4 applies to any securities lending agreement entered into or executed by Party A and Party B prior to the date of this Agreement, including, without limitation, any Global Master Securities Lending Agreement (2000 version or 2010 version). The Parties agree that this Agreement supersedes and replaces any such prior agreement and such prior agreement shall cease to apply.

12. AUTOMATION

Paragraph 27.5 does apply

13. AMENDMENTS AND SUPPLEMENTAL TERMS

The following provisions shall amend or supplement, as applicable, the terms of the Agreement:

- (a) The following additional provisions shall be inserted into Paragraph 1 of the Agreement as Paragraph 1.4:
 - "1.4 References in this Agreement to "it" or "its" will be replaced with the words "he or she", "him or her" or "his or her" as appropriate in so far as they apply to Party B and shall (for the avoidance of doubt) include, where appropriate, any executor or administrator of the estate thereof."
- (b) The following additional provisions shall be inserted into Paragraph 10.1 of the Agreement:

The word "or" shall be added at the end of Paragraph 10.1 and the following paragraphs shall be inserted in paragraph 10.1 immediately after Paragraph 10.1(i):

- "(j) The Charge on Assets terminates, is or becomes unenforceable for any reason or is withdrawn for any reason; or
- (k) Party B is convicted of a criminal offence (1) in Hong Kong other than one for which a fixed penalty is prescribed under the Fixed Penalty (Traffic Contraventions) Ordinance (Cap. 237 of the laws of Hong Kong) or the Fixed Penalty (Criminal Proceedings) Ordinance (Cap. 240 of the laws of Hong Kong); (2) in the PRC under the Criminal Law of the People's Republic of China, as amended from time to time, whether the criminal penalties are imposed as fines or imprisonment; or (3) in any other jurisdiction other than an offence committed in such other jurisdiction which is dealt with by a procedure substantially similar to that applicable to a fixed penalty offence in Hong Kong; or
- (l) Any event or series of events occurs and is continuing which, in the opinion of Party A (acting in its sole and absolute discretion), has or is reasonably likely to have a Material Adverse Effect;

Where "Material Adverse Effect" means a material adverse effect on:

- (1) the business, assets, financial condition or credit worthiness of Party B;
- (2) the ability of Party B to perform its obligations under this Agreement; or
- (3) any right or remedy of Party A in respect of this Agreement; or
- (m) Party B fails to provide Party A with adequate assurance of its ability to perform its outstanding obligations under this Agreement or Charge on Assets (as applicable) within one Local Business Day after a request is made by Party A, when Party A has reasonable grounds for making such request; or
- (n) The occurrence of a default or any other similar event or the exercise of any right of liquidation, acceleration, early termination, enforcement, set-off, lien, right of retention, combination of accounts, appropriation or any other remedy, in each case as described in any transaction or agreement between (a) Party B or any Affiliate of Party B; and (b) Party A or any Affiliate of Party A (including but not limited to the Client Agreement for Securities Trading and Agreement for Securities Margin Trading), as determined by Party A in its sole and absolute discretion; or
- (o) Party A determines, acting in its sole and absolute discretion, that the credit-worthiness of Party B becomes materially weaker; or
- (p) Party B dies or, in the reasonable opinion of Party A, becomes incapable in law of independently managing all or part of its assets, income and liabilities in any relevant jurisdiction (whether by reason of any physical or mental disability or other incapacity or incompetence or otherwise); or

- (q) A default, potential event of default, event of default, termination event (however described) occurs or is declared in respect of Party B or any Affiliate of Party B under any transaction or agreement (whether or not arising under the Agreement) between (a) Party B or Affiliate of Party B and (b) Party A or any Affiliate of Party A; or
- (r) Party B fails to comply with or perform any agreement or obligation to be complied with or performed by it in accordance with any agreement (including but not limited to Charge on Assets) entered into between Party A (and/or its Affiliates) and Party B; or
- (s) A representation made or repeated or deemed to have been made or repeated by Party B in any agreement (including but not limited to Charge on Assets) entered into between Party A (and/or its Affiliates) and Party B proves to have been incorrect or misleading when made or repeated or deemed to have been made or repeated; or
- (t) In relation to Party B or Affiliate of Party B, any indebtedness, liability or obligation (whether cash, securities or otherwise) is not paid, met or delivered at its stated maturity (or within any applicable grace period) or, becomes due prior to its stated maturity or if payable or repayable on demand, when so demanded."
- (c) The following additional definitions shall be included in Paragraph 2.1:

"Affiliate" means, (i) with respect to Party A, any entity controlled, directly or indirectly by Party A, any entity that controls, directly or indirectly, Party A or any entity directly or indirectly under common control with Party A; and (ii) with respect of Party B, any entity controlled directly or indirectly by Party B and any other person identified for that purpose by notice in writing from Party A to Party B from time to time. For this purpose, "control" of any entity or person means ownership of a majority of the voting power of the entity or person.

"SFO" means the Securities and Futures Ordinance (Cap. 571 of the laws of Hong Kong).

(d) Amended definitions

The definition of "**Act of Insolvency**" in paragraph 2.1 of this Agreement is amended by adding the following words to the end of the definition:

"or,

(g) with respect to Party B:

- (1) Party B is, or is deemed for the purposes of any applicable law to be, unable to pay its debts as they fall due or is declared or is deemed to be bankrupt, suspends making payments on any of its debts or by reason of actual or anticipated financial difficulties, commences negotiations with one or more of its creditors (excluding Party A in its capacity as Lender or Borrower under this Agreement) with a view to rescheduling any of its indebtedness;
 - (2) Party B admits its inability to pay its debts as they fall due;
 - (3) Party B commits an act of bankruptcy;
- (4) Party B declares himself or herself bankrupt (if applicable), or applies for individual voluntary arrangement or its equivalent in any jurisdiction; or
- (5) Party B served with a statutory demand under Section 6A of the Bankruptcy Ordinance (Cap. 6 of the laws of Hong Kong) (or any analogous document in any jurisdiction

whatsoever), named as a dishonest judgement debtor or served with a restriction order on high consumptions under the Several Provisions on Information Disclosure concerning the List of Dishonest Judgment Debtors as amended from time to time or its equivalent in any other jurisdictions."

(e) Margin Threshold

Paragraph 5.4 of the Agreement is amended by adding the following at the end of sub-paragraph (d), a new sub-paragraph (e):

"Notwithstanding anything to the contrary in this Agreement, neither party may require the other party to pay or deliver further Collateral or to redeliver Equivalent Collateral under this Agreement if the Market Value of the Collateral or Equivalent Collateral which is required to be paid or delivered amounts to zero ("0") (*Margin Threshold*). The Margin Threshold shall not apply in the event that there are no outstanding obligations with respect to both parties under any Loan."

- (f) Pursuant to the terms of paragraph 1.2 of the Agreement, Borrower and Lender agree to be governed by the supplemental terms and conditions stated herein. To the extent that any provisions in these supplemental terms and conditions are in conflict with provisions contained in the Agreement, the provisions contained in these supplemental terms and conditions shall prevail.
- (g) Party B shall deliver to Party A the evidence of signing authority including specimen signatures promptly after the executing this Agreement. Party B shall deliver (i) evidence of the consent to this Agreement of any spouse, partner, joint tenant or tenant-in-common of Party B or other person with whom Party B owns property jointly, to the extent that such person is obliged or such property is available to meet claims by Party B under this Agreement and evidence that in giving such consent such person has received independent legal advice and (ii) a copy of his/her identity card or passport (as applicable).
- (h) For the purposes of Paragraphs 5.4 and 5.5 of the Agreement, the parties agree that the amount or Market Value of Income payable in respect of Non-Cash Collateral or (as the case may be) Equivalent Securities shall not be taken into account in the calculations carried out in Paragraph 5.4(b), 5.4(c), 5.5(b) or 5.5(c).
- (i) Where any Loan consists of Hong Kong stocks (as such term is defined under the Hong Kong Stamp Duty Ordinance (the "SDO")), the Borrower agrees to the following:
 - 1. it shall be subject to and responsible for compliance with all applicable provisions and requirements of the SDO which shall include:
 - (1) the timely registration of this Agreement with the Collector of Stamp Revenue as appointed under the SDO (the "Collector") in accordance with Section 19(12A) of the SDO in Hong Kong including the provision of:
 - (i) two certified true copies of this Agreement;
 - (ii) such fees and duties as may be specified from time to time under the SDO; and
 - (iii) such other documents, particulars and information as the Collector may require (including a duly completed "Stock Borrowing and Lending Agreement Registration Form" in duplicate),

at any time after the execution of this Agreement and in any event before the expiry of thirty (30) days after the first stock borrowing of Hong Kong stocks is effected;

- (2) all filing, record-keeping and six-monthly reporting obligations as may be required under the SDO; and
- (3) any other acts and things as may be required by the Collector
- 2. it warrants and undertakes on a continuing basis that it shall only borrow Hong Kong stock for one or more "specified purposes" as defined under the SDO; and
- 3. it shall indemnify the Lender in respect of any costs, fees, penalties, liability or loss incurred by the lending party as a result of or in connection with:
 - (a) the Borrower's failure to comply with the SDO requirements referred to in paragraph (1) above; or
 - (b) any breach by the Borrower of its undertakings set out in paragraph (2) above.
- (j) Party B will be deemed to represent to Party A on the date on which he/she enters into a Loan that:
 - (i) He/she is acting for his/her own account, and he/she has made his/her own independent decisions to enter into that Loan and as to whether that Loan is appropriate or proper for him/her, having regard to his/her financial position, investment objectives and other personal circumstances, and based upon his/her own judgement and upon advice from such advisers as he/she has deemed necessary. He/she acknowledges that information and explanations related to the terms and conditions of a Loan will not be considered investment advice or a recommendation to enter into that Loan. No communication (written or oral) received from Party A or its Affiliates will be deemed to be an assurance or guarantee as to the expected results of that Loan;
 - (ii) He/she is capable of assessing the merits of and understanding (on his/her own behalf or through independent professional advice which he/she has been advised to obtain before entering into this Agreement or that Loan), and understands and accepts, the terms, conditions and risks of that Loan (including the risk and extent of loss which could be substantial in certain circumstances), which have been drawn to his/her attention. He/she is also capable of assuming, and assumes, the risks of that Loan (including the risk of loss which could be substantial in certain circumstances);
 - (iii) Neither Party A nor any of Party A's Affiliates is acting as a fiduciary for him/her in respect of that Loan; and
 - (iv) He/she is entering into this Agreement and each Loan as principal and not as agent of any person.
- (k) Paragraph 23 shall be deleted in its entirety and replaced with the following:

"GOVERNING LAW AND JURISDICTION

- 23.1 This Agreement and any non-contractual obligations arising out of or in connection with this Agreement shall be governed by, and shall be construed in accordance with, English law.
- 23.2 The courts of the Hong Kong SAR have exclusive jurisdiction to hear and decide any suit, action or proceedings, and to settle any disputes or any non-contractual obligation which may arise out of or in connection with this Agreement (respectively, *Proceedings* and *Disputes*) and, for these purposes, each Party irrevocably submits to the jurisdiction of the courts of the Hong Kong SAR.
- 23.3 Each Party irrevocably waives any objection which it might at any time have to the courts of the Hong Kong SAR being nominated as the forum to hear and decide any Proceedings and to settle any Disputes and agrees not to claim that the courts of Hong Kong SAR are not a convenient or appropriate forum.
- 23.4 Each Party hereby respectively appoints the person or entity identified in paragraph 7 of the Schedule pertaining to the relevant Party as its agent to receive on its behalf service of process in the courts of the Hong Kong SAR. If such an agent ceases to be an agent of a Party, the relevant Party shall promptly appoint, and notify the other Party of the identity of its new agent in the Hong Kong SAR.

(1) Personal Data (Privacy) Ordinance

- (i) Personal data held by Party A relating to Party B, Party B's agent(s) or dependant(s) (if any) may be used for the purposes of this Agreement and any Loan and in relation to the distribution of research, enforcement against Party B, risk assessment, compliance with regulatory requirements to know Party B and to carry out due diligence to assess Party B's investment suitability and for any other directly related purposes and will be kept confidential, but Party A may provide such information to:
 - (1) any other branch or office of Party A or its Affiliates;
 - (2) any agent, contractor, other service provider or other person (including any credit agency) who provides administrative, telecommunications, computer or other facilities or services to Party A in connection with the operation of Party A's business or the operation of the Agreement notwithstanding that such agent, contractor, service provider or person may be outside Hong Kong who may not be subject to the Personal Data (Privacy) Ordinance (Cap. 486 of the laws of Hong Kong);
 - (3) any person who owes a duty of confidentiality to Party A or its Affiliates which have undertaken to keep such information confidential;
 - (4) any financial institution with which Party B has or proposes to have dealings;
 - (5) any actual or proposed assignee of Party A or participant or sub-participant or transferee of Party A's rights in respect of Party B; or
 - (6) any regulatory authority or judicial body as and when requested to do so.
- (ii) Party B hereby agrees that Party A may transfer personal data (including to a place outside Hong Kong) in certain circumstances as specified in this Paragraph (1).

(iii) Party B understands that Party B has the right pursuant to the Personal Data (Privacy) Ordinance (Cap. 486 of the laws of Hong Kong) to require Party A (at no cost to Party B) to access and/or correct Party B's personal data held by Party A. Such requests should be made in writing and the person to whom requests for access to, or correction of, data, or for information regarding policies and practices and kinds of data held, should be addressed is the Data Protection Officer of Guotai Junan Securities (Hong Kong) Limited at 27/F, Low Block, Grand Millennium Plaza, 181 Queen's Road Central, Hong Kong.

(m) Additional Representations of Party B

Party B represents and warrants to Party A (which representations shall be deemed to be repeated on each day on which a Loan is entered into and on each day on which Securities or Collateral are to be transferred under any transaction under this Agreement) that:

- (i) He or she is aware of, fully understands and agrees that he or she will be responsible for complying with and has met any disclosure, reporting, filing and/or registration obligation to which he or she is subject, as required by applicable law and regulation and accounting and regulatory standards, including, without limitation to, all of his or her obligations under Part XV (*Disclosure of Interests*) of the SFO and the Loans have and will be properly accounted for and reported to his or her auditors in accordance with such standards;
- (ii) He or she is of full age and sound mind and has full capacity to enter into this Agreement. He or she is single or, if married, his or her spouse is aware of the execution by him or her of this Agreement and any documentation relating to this Agreement to which he or she is a party, and the transactions and matters contemplated hereby and thereby, and his or her spouse has not raised any objections thereto;
- (iii) He or she is not a substantial shareholder, chief executive, director and/or insider in respect of any of the Securities or Collateral (including any new securities that may be credited to its securities account from time to time);
- (iv) The funds and assets in relation to the performance of this Agreement are derived from legitimate sources outside the People's Republic of China (which for these purposes, excludes Hong Kong, Macau and Taiwan) ("PRC") and are used in a compliant manner that does not constitute a breach or circumvention of the applicable foreign exchange rules and regulations of the PRC;
- (v) Party A does not offer to sell or solicit an offer to buy any securities or investment products under this Agreement in any relevant jurisdiction;
- (vi) He or she has and will at all times fully comply with all applicable laws and regulations in each jurisdiction with regard to each Loan, including without limitation, the Securities and Futures Ordinance (Cap. 571 of the laws of Hong Kong), the Hong Kong Codes on Takeovers and Mergers and Share Buy-backs, and the Rules Governing the Listing of Securities on the Stock Exchange of Hong Kong Limited; and
- (vii) He or she is not a consumer or dealing with Party A as a consumer for the purposes of any or all of the Sale of Goods Ordinance (Cap. 26 of the laws of Hong Kong), Control of Exemption Clauses Ordinance (Cap. 71 of the laws of Hong Kong), Supply of Services (Implied Terms) Ordinance (Cap. 457 of the laws of Hong Kong) or the Unconscionable Contracts Ordinance (Cap. 458 of the laws of Hong Kong).

(n) Charge on Assets

The	Parties	acknowledge	the	existence	of	a	charge	on	assets	("Charge	on	Assets")	dated
					,	ex	recuted	by I	Party B	in favour	of l	Party A.	

(o) Where Party A is the Borrower in respect a Loan, the Collateral with respect to that Loan to be delivered by Party A to Party B shall be subject to a charge created by Party B in favour of Party A pursuant to the Charge on Assets and that Collateral shall form part of the Charged Assets (such term as defined in the Charge on Assets). Party B agrees and acknowledges that, during the course of that Loan, Party B may not withdraw the Collateral and Party B will not be granted any purchasing power or trading limit in respect of that Collateral.

(p) Risk Disclosure

Party B confirms that he/she has reviewed the risk disclosures contained in the Risk Disclosure Statement Annex to this Agreement. Party B acknowledges and accepts all of the risks arising from Loans entered into under this Agreement, some of which have been described and detailed in the Risk Disclosure Statement Annex.

- (q) Accurate Information: Party B confirms that the information provided in this Agreement and/or other information collection documents is complete, accurate and current. Party B acknowledges the importance to provide complete, accurate and current information to Party A as Party A will rely on such information to understand the Party B's personal background, financial situation and other necessary information for Party A to provide stock lending or stock borrowing services or products to Party B. Party B undertakes to inform Party A immediately if there is any change to that information. It is Party B's responsibility to ensure information accuracy and to notify Party A immediately with regard to any discrepancies. Party A also undertakes to inform Party B of any material change of Party A's name, address, registration status, services, remuneration, fees, rates and margin provisions in relation to the provision of stock lending or stock borrowing services or products to Party B.
- (r) If Party A ("we"), solicit the sale of or recommend any financial product to Party B ("you"), the financial product must be reasonably suitable for you having regard to your financial situation, investment experience and investment objectives. No other provision of this Agreement or any other document we may ask you to sign and no statement we may ask you to make derogates from this clause.

(s) Joint Account

- (i) If there are two or more persons acting as Party B under this Agreement (the "**Persons**", each a "**Person**"), the expression "Party B" shall mean all or each of them and their obligations and liabilities under this Agreement shall be joint and several.
- (ii) Any communication under this Agreement served on any one Person shall be deemed to have been served on all of them. Each Person agrees to be bound by this Agreement notwithstanding that any others of the Persons which were intended to sign or be bound may not do so or be effectually bound and notwithstanding that this Agreement may be determined or become invalid or unenforceable against any one or more of such Persons whether or not the deficiency is known to Party A. Party A shall be entitled to release any one or more of the Persons from this Agreement, to compound with or otherwise vary or agree to vary the liability of, or to grant time or other indulgence to, or make other arrangements with, any one or more of the Persons, without prejudicing or affecting Party A's rights, powers and remedies by Party A against any other Person.
- (iii) The occurrence of an Event of Default to any Person shall be construed as an Event of Default, as applicable, occurring with respect to each Person acting as Party B.

(t)	English/Chinese Version: Party B ("you") confirm that you have read the English or Chinese version of this Agreement and that the contents of this Agreement have been fully explained to you in a language which you understands, and that you accept this Agreement in its entirety.

EXECUTED by the **PARTIES**

SIGNED by)
)
)
)
)

duly authorised for and on behalf of Guotai Junan Securities (Hong Kong) Limited I have carefully read and understood this Agreement, including the Risk Disclosure Statement Annex, and that the contents of this Agreement, including the Risk Disclosure Statement Annex, was provided in a language of my choice. I accept all the terms of this Agreement and agree to be bound by it as amended and supplemented from time to time.

I acknowledge that I have been invited to read and ask questions about this Agreement, including the Risk Disclosure Statement Annex and I am invited to take independent advice if I wish. I am satisfied with the explanation and clarification provided to me. I declare that I am willing and capable of fulfilling the duties and obligations as specified in this Agreement.

SIGNED by	
)
holder of	
number	

SIGNED and DECLARED by Licensed Representative who has provided and explained to Party B the content of this Agreement including the Risk Disclosure Statement Annex in the language of Party B's choice; and has invited Party B to read the Risk Disclosure Statement Annex, to ask questions and to take independent advice if Party B wish.

Signature of Licensed Representative
Full name of Licensed Representative
CE Number of Licensed Representative
Date

Risk Disclosure Statement Annex

Party B ("you") should ensure that you have read the terms and conditions of this Agreement and this Risk Disclosure Statement Annex carefully and sought advice from financial, legal and tax advisers (amongst others) to consider and understand the risks arising under this Agreement. Do not sign this Agreement or enter into any Loan unless you fully understand and are willing to assume the associated risks. This Agreement or any Loan contemplated under this Agreement may not be suitable for everyone and you should consider whether this Agreement or any Loan contemplated under this Agreement is suitable for you in light of your own financial position, risk appetite, investment experience and investment objectives. Pursuant to the SFO, the Investor Compensation Fund only covers products trading in Hong Kong's recognised securities market. The Loans contemplated under this Agreement are not traded on any markets operated by Hong Kong Exchanges and Clearing Limited or any other stock exchanges and therefore any loss you may sustain by reason of a default by any Securities and Futures Commission licensed or registered person are not covered by the Investor Compensation Fund.

(a) Credit Risk

Where Party A ("we") have borrowed Loaned Securities from you, you will not retain any proprietary interest in such Loaned Securities which have been delivered to us. Instead, you will have a contractual right to receive Equivalent Loaned Securities from us at the end of the Loan. In the event that we become insolvent, you will become one of our unsecured creditors with only an unsecured contractual claim against us to claim the value of the Equivalent Loaned Securities, but will have no right to the return of the Loaned Securities or the physical delivery of any Equivalent Loaned Securities. What you will receive in respect of any debt owed to you by us will depend on the assets available to the creditors after any insolvency procedures have been completed, and the debt may not be repaid in full, or at all. In the worst-case scenario, you may not receive any Equivalent Loaned Securities and/or payment of any lending fees or other sums owed to you.

Where you are the Lender under a Loan, you will be exposed to the credit risk of us for the duration of such Loan in respect of the market value of the Loaned Securities. A default by us could result in the loss of the Loaned Securities. In the worst-case scenario, you may not receive any Equivalent Loaned Securities and/or payment of any fees or other sums owed to you.

Where you are a Borrower under a Loan, you will be exposed to the credit risk of us on the return of collateral. In the event that we should default on our obligations or become insolvent, you may not receive back the collateral. You may monetise the Loaned Securities to minimise loss.

(b) Early Termination of Loans

A Termination Date may be specified on the occurrence of any Event of Default to either you or us, which includes (but is not limited to) any failure to pay amounts due under any Loan, bankruptcy/insolvency, misrepresentation and breach of any obligation under this Agreement. If a Termination Date occurs, all Loans will be accelerated and terminated, and all payment and delivery obligations between us will be valued in accordance with the Agreement, with a net amount being payable by either you to us or us to you. You will not be entitled to the return of any Loaned Securities in this case.

(c) Conflicts of Interest

We, or a member of our group, may from time to time, as principal or agent, have positions in, may buy or sell, or make a market in the Loaned Securities. We have policies and procedures designed to minimise the risk that officers and employees are influenced by any conflicting interest or duty and that confidential information is improperly disclosed or made available. Nevertheless, our general trading and hedging activities may affect the market value of the Equivalent Loaned Securities, which may have an impact on the Loans.

(d) Tax

You will be responsible for any and all taxes of any jurisdiction or governmental or regulatory authority that may be applicable to Loans. Additional amounts may not be paid to you to reimburse you for any tax, assessment or charge required to be withheld from payments. You acknowledge that we do not have any responsibility to you to provide any advice or to take any steps to minimize or estimate any tax that may be suffered by you in respect of the Loans.

(e) Currency Risk

Where the Loaned Securities are denominated in a currency which is different from the currency in which the relevant securities account is held, you will be exposed to fluctuations in foreign exchange rates, which may at times move against you and cause losses.

(f) Delivery / Settlement Risk

The assumption that Securities can be delivered may be affected by disruption events beyond the control of you and/or us.

If we fail or delay to return the Equivalent Loaned Securities to you, you may suffer consequential losses and/or loss of profit as a result. You may not be able to claim any costs, expenses, losses and damages or any sums in respect of such consequential loss or loss of profit (whether arising directly or indirectly) from us.

(g) Potential Unlimited Liability

Upon entering into this Agreement, you undertake to indemnify us in certain situations, including without limitation, against any liability arising in respect thereof as a result of your failure to pay Tax. Such liability may be unlimited; for instance, up to ten times of the Tax may be payable if the Tax was not paid to the relevant regulator within the prescribed time limit.

(h) Volatility Risk

In adverse market conditions, volatility can increase and this will lead to greater collateralisation calls resulting in a larger placement/receipt of cash collateral. In normal market conditions, volatility in securities markets will vary between individual securities.

In more volatile market conditions clients may be subject to a wider spread with regards to pricing available.